Age- and sex-specific reference values of estimated glomerular filtration rate for European adults



Megan E. Astley^{1,2}, Nicholas C. Chesnaye^{1,3}, Stein Hallan^{4,5}, Giovanni Gambaro⁶, Alberto Ortiz⁷, **OPEN** Juan-Jesus Carrero^{8,9}, Natalie Ebert¹⁰, Bjørn Odvar Eriksen^{11,12}, Anne-Laure Faucon^{8,13}, Pietro Manuel Ferraro¹⁴, Olafur S. Indridason¹⁵, Till Ittermann¹⁶, Arnar J. Jonsson^{17,18}, Knut Asbjørn Rise Langlo^{4,5}, Toralf Melsom^{11,12}, Elke Schaeffner¹⁰, Sylvia Stracke¹⁹, Vianda S. Stel^{1,3} and Kitty J. Jager^{3,20}

¹ERA Registry, Department of Medical Informatics, Amsterdam UMC location University of Amsterdam, Amsterdam, the Netherlands; ²Health Behaviours and Chronic Diseases and Methodology, Amsterdam Public Health Research Institute, Amsterdam, the Netherlands; 3 Quality of Care, Amsterdam Public Health Research Institute, Amsterdam, the Netherlands; 4 Department of Clinical and Molecular Medicine, Faculty of Medicine and Health Sciences, Norwegian University of Science and Technology, Trondheim, Norway; ⁵Department of Nephrology, St. Olavs Hospital, Trondheim, Norway; ⁶Department of Medicine, Division of Nephrology and Dialysis, University of Verona, Verona, Italy; ⁷Department of Nephrology and Hypertension, IIS–Fundacion Jimenez Diaz UAM, Madrid, Spain; ⁸Department of Medical Epidemiology and Biostatistics, Karolinska Institutet, Stockholm, Sweden; ⁹Division of Nephrology, Department of Clinical Sciences, Danderyd Hospital, Danderyd, Sweden; ¹⁰Institute of Public Health, Charité Universitätsmedizin Berlin, Berlin, Germany; ¹¹Section of Nephrology, Clinic of Internal Medicine, University Hospital of North Norway, Tromsø, Norway; ¹²Metabolic and Renal Research Group, UIT The Arctic University of Norway, Tromsø, Norway; ¹³Department of Clinical Epidemiology, Centre for Research in Epidemiology and Population Health, INSERM U1018, Paris-Saclay University, Paris, France; ¹⁴Section of Nephrology, Department of Medicine, Università degli Studi di Verona, Verona, Italy; ¹⁵Division of Nephrology, Landspitali University Hospital, Reykjavik, Iceland; ¹⁶Institute for Community Medicine - SHIP Clinical Epidemiological Research, University Medicine Greifswald, Greifswald, Germany; ¹⁷Internal Medicine Services, Landspitali University Hospital, Reykjavik, Iceland; ¹⁸Faculty of Medicine, University of Iceland, Reykjavik, Iceland; ¹⁹Division of Nephrology, Internal Medicine A, University Medicine Greifswald, Greifswald, Germany; and ²⁰Amsterdam UMC location University of Amsterdam, Medical Informatics, Amsterdam, the Netherlands

Kidney function, often assessed by estimated glomerular filtration rate (eGFR), declines naturally with age. However, there is a lack of eGFR reference values to describe normal and abnormal values for a specific age. The European Chronic Kidney Disease Burden Consortium is comprised of nine participating general population-based studies from seven European countries which provides European ageand sex-specific eGFR reference values in healthy adults using the European Kidney Function Consortium (EKFC) equation. Of 2,572,020 individuals, 1,535,253 (60%) were considered healthy, of which 45% were men. Ages ranged from 18 to 105 years old in men and 18 to 107 years old in women with a median age of 43 years in both sexes. At age 20 in men, the 5th, 50th and 95th eGFR percentiles were 78 ml/min per 1.73 m², 99 ml/min per 1.73 m², and 119 ml/min per 1.73 m². In 20-year-old women this was 81 ml/min per 1.73 m², 101 ml/min per 1.73 m², and 121 ml/min per 1.73 m². Consequently, in men aged 80 years old, the 5th, 50th and 95th eGFR percentiles were 49 ml/min per 1.73 m², 66 ml/min per 1.73 m², and 84 ml/min per 1.73 m². In 80 year old women this was 46 ml/min per 1.73 m², 63 ml/min per 1.73 m², and 81 ml/min per 1.73 m². Overall, our study shows that eGFR is not preserved with ageing in healthy

arch 2025

individuals and these eGFR reference values can help determine abnormal and normal kidney function across the age range.

Kidney International (2025) **107,** 1076–1087; https://doi.org/10.1016/ j.kint.2025.02.025

KEYWORDS: estimated glomerular filtration rate; general population; kidney function; reference values

Copyright © 2025, International Society of Nephrology. Published by Elsevier Inc. This is an open access article under the CC BY license (http:// creativecommons.org/licenses/by/4.0/).

n estimated 10% of the world has chronic kidney disease (CKD). The Kidney Disease: Improving Global Outcomes (KDIGO) definition is most commonly used to diagnose CKD stages.¹ Whatever the etiology, it classifies CKD into stages based on albuminuria and/or glomerular filtration rate values, from the less severe stage G1 to the more advanced stage G5, preluding dialysis or kidney transplantation. However, in the literature, a frequently used definition of CKD (actually corresponding to KDIGO stages 3-5) is an estimated glomerular filtration rate (eGFR) <60 ml/min per 1.73 m² present for a minimum of 3 consecutive months, independent of age and sex. This definition does not take into account kidney function decline that occurs with aging, even in healthy individuals, resulting in claims that CKD is being underdiagnosed in younger people and overdiagnosed in older people.² To account for age in the CKD definition, it has been suggested to use age-group-specific

Correspondence: Megan Astley, Amsterdam UMC location University of Amsterdam, Department of Medical Informatics, Meibergdreef 9, 1105 AZ Amsterdam, the Netherlands. E-mail: m.e.astley@amsterdamumc.nl

Received 10 September 2024; revised 4 February 2025; accepted 17 February 2025; published online 22 March 2025

Lay Summary

Kidney function, often assessed by estimated glomerular filtration rate (eGFR), naturally declines with age. Despite this, current criteria for diagnosing chronic kidney disease (CKD) do not account for this age-related decline. Overlooking age-related changes in kidney function may hinder the assessment, diagnosis, management, and treatment of kidney disease. Establishing age-specific eGFR reference values could be used to determine if an individual's eGFR is abnormal for his/her age and help refine the current standard to diagnose CKD. Using data collected on 1.5 million healthy individuals from Europe, we describe the distribution of eGFR values in adults aged 18 to 100 years. Our results show that the definition of abnormal kidney function varies by age and provide clinically relevant information that may contribute to improving practices and policies for the surveillance, identification, and appropriate management of abnormal kidney function and CKD.

eGFR thresholds or a continuous eGFR threshold based on the age- and sex-specific distribution of eGFR in healthy adults.²

A prerequisite for the development of a continuous threshold is the ability to distinguish between normal and abnormal kidney function, thus requiring the establishment of eGFR reference values.³ Reference values are the range of values considered normal for a particular population and are used in other medical fields, such as for pediatric growth charts⁴ and lung spirometry.⁵ However, eGFR reference values have been difficult to define. Delanaye *et al.* outlined the most common obstacles that studies face, including small sample sizes, unrepresentative samples, poorly defined health criteria, and insufficient age ranges.³

To date, several studies described eGFR reference values in kidney donors.⁶⁻¹¹ These studies include highly selected groups of people and were mostly of small sample sizes. Other studies in healthy individuals sampled from European general populations have relatively small sample sizes, are singlecountry studies,^{12,13} are likely unrepresentative of the larger European population, or use outdated eGFR equations.^{14,15} A more recent study from Eriksen et al. provided reference values for measured glomerular filtration rate (mGFR) using samples from 3 European general population studies, but the age range was limited to individuals aged >50 years.¹⁶ Regardless of the strengths of the aforementioned studies, fundamental differences between studies make it difficult to combine results and obtain eGFR reference values from a sufficiently large sample representative of the general population, and covering a full age range, for use in Europe.

Therefore, this study aims to establish European eGFR reference values by describing the distribution of eGFR values across a full age range by sex, in >1.5 million healthy

individuals from a multinational, contemporary, adult European general population.

METHODS

Study selection

General population studies were identified by a literature database search, outreach, and from a previous study of the European CKD Burden Consortium.¹⁷ Studies were included if they were designed to select a representative sample of the target population. Studies needed to have a minimum sample size of 2000 individuals and collected data after 1999 on adults aged \geq 18 years with isotope dilution mass spectrometry traceable serum creatinine.

Data collection

Studies provided an individual-level data set containing data on sex, age at moment of data collection, serum creatinine, and if the individual was considered healthy, based on the definition described further in the Methods. Missingness of data on comorbidities, medication use, and lifestyle-related risk factors used to define the healthy individual was also provided. Additional patient characteristics were collected as aggregated data. Date of data collection could be at any time between January 2000 and April 2024. Data collected were either self-reported or obtained through medical records using International Classification of Diseases, Tenth Revision (ICD-10), codes. If ICD-10 codes were used to identify comorbid conditions, we assumed that the absence of an ICD-10 code indicated the absence of the considered comorbidity. Smoking was determined using self-reporting and not by *ICD-10* codes. We realize there is a difference between gender and sex, but we will use the term sex throughout this article as we expect differences observed between healthy men and women to be primarily a result of biological differences.

eGFR

eGFR was calculated using the European Kidney Function Consortium (EKFC) equation, which is listed as a validated equation by KDIGO.^{18,19} Serum creatinine was determined by Jaffe or enzymatic assays and calibrated to isotope dilution mass spectrometry. The EKFC equation and Q values are described in detail in the Supplementary Methods.¹⁸ We repeated the analysis using the Revised Lund-Malmö (RLM) equation,²⁰ the Chronic Kidney Disease Epidemiology Collaboration (CKD-EPI)₂₀₀₉ equation,²¹ and the CKD-EPI₂₀₂₁ equation.²² Race was considered as non-Black for all individuals in the calculation of eGFR using the CKD-EPI₂₀₀₉ equation, as most studies had a predominantly White study population, did not collect data on race, or were not allowed to collect data on race. Data on cystatin C were not collected, and only serum creatinine–based eGFR is reported.

Definition of healthy individuals

To minimize inclusion of individuals who may be at risk or have kidney damage, individuals were considered healthy if they met the following criteria at the time of data collection:

(i) no history of myocardial infarction, (ii) no history of angina pectoris, (iii) no history of heart failure, (iv) no history of coronary artery disease, (v) no history of hypertension, (vi) no history of stroke, (vii) no history of cancer, (viii) no history of diabetes, (ix) no history of kidney disease, (x) body mass index \leq 30 kg/m², (xi) never smoked, (xii) no use of lipid-lowering medication or cardiac glycosides, (xiii) urine albumin-to-creatinine ratio \leq 30 mg/g, and (xiv) no use of angiotensin-converting enzyme inhibitors or angiotensin receptor blockers. These criteria to define healthy were based on, and slightly modified from, a healthy definition described by Eriksen et al.¹⁶ Missingness in the variables used to define the healthy population is described in Supplementary Table S1 for each study. Collection methods among cohorts varied, with some variables missing at the cohort level. If the variable was not collected by a study, we considered the individuals to have no history or presence of that exclusion criteria rather than consider it missing. Individuals missing data were not included in the analysis.

Statistical analysis

Normally distributed continuous variables were presented as means with SD. Skewed continuous variables were presented as medians with interquartile range. Categorical variables were presented as frequencies with percentages.

The relationship between eGFR and age was explored using a generalized additive model for location, scale, and shape, which allows the distribution of eGFR to be modeled nonlinearly across age by sex with smoothing functions and random effects for study, which accounts for the clustering of observations within each study.²³ The generalized additive model for location, scale, and shape also allows the variance of eGFR to be modeled as a separate function of age. Model specifications are described in the Supplementary Methods. Analyses were performed using the gamlss package version 5.4-20 in R version 4.4.1.^{24,25} Model comparisons were performed using Akaike information criterion to decide which smoothing functions should be used. Graphically, we present the median eGFR distribution with the 5th and 95th percentiles by sex for those aged 18 to 100 years.

Sensitivity analyses

To assess for the potential influence of cohorts with significantly larger sample sizes on our results, we repeated the main analysis on a subset of data by randomly selecting 1000 healthy individuals from each cohort where available, or all healthy individuals from cohorts with <1000 healthy individuals. eGFR calculated using the EKFC equation was compared with mGFR in a subset of individuals using bias, precision, and accuracy.

RESULTS

Participating cohorts' characteristics

Nine population-based cohorts from Iceland, Italy, Germany, the Netherlands, Norway, Sweden, and the United Kingdom participated (Table 1^{26-35}). Most individuals came from the

Stockholm CREAtinine Measurement (SCREAM) study (72%), the UK Biobank (14%), the Iceland CKD study (9%), and the Lifelines cohort (3%), whereas the remaining studies contributed <1% of the analytic cohort. Most studies recruited participants from general practitioner lists or population registers (Table 1 and Supplementary Table S2). Response rates ranged from 54% to 69%, with the exception of the UK Biobank (5.5%) and The Berlin Initiative Study (BIS; 8.1%), which still demonstrated a high agreement with regard to the distribution and frequency of chronic diseases when compared with its source population.²⁸ The smallest cohort had 2068 individuals (<1%; the BIS study, Germany) and the largest >1.6 million individuals (72%; SCREAM, Sweden). Six cohorts covered a relatively wide age spectrum, whereas the 3 remaining cohorts did not include individuals aged <40 years. The collection of the 14 variables used to define healthy ranged from as low as 10 variables to all 14 (Supplementary Table S1). Figure 1 describes the exclusion of individuals based on missing data and health status. Missingness in the variables used to define the healthy population was generally low (<3%), with some exceptions (Supplementary Table S1). Of 2,572,020 individuals, 2,504,048 (97%) had data to decide health status. Of these, 1,535,253 (60%) were classed as healthy, of which 45% were men. A similar proportion of women (63%) were categorized as healthy as men (60%). Density plots reflecting the distribution of healthy and not healthy individuals by sex, age, and study are presented in Supplementary Figure S1.

Population characteristics of healthy individuals by sex

Population characteristics for healthy individuals in each study are presented in Table 2, and by sex in Supplementary Table S3. Median age was 43 (interquartile range, 23) years in men and 42 (interquartile range, 24) years in women. Median eGFR using the EKFC equation was 95 ml/min per 1.73 m² in men and 96 ml/min per 1.73 m² in women.

eGFR distribution over age in healthy individuals by sex

The median eGFR distribution with 5th and 95th percentiles for men and women using the creatinine-based EKFC equation (EKFC_{crea}) is presented in Figure 2. eGFR distributions by 5-year age intervals by sex are presented in Table 3. eGFR was relatively stable until the age of 40 years, after which point eGFR decreased with age, reflecting the structure of the EKFC equation, which accounts for age-related changes in kidney function. Women had eGFR comparable to men before the age of 60 years, but a lower eGFR thereafter. There was a significant sex-age interaction (P < 0.001). The 5th and 95th percentiles of eGFR were on average 22 ml/min per 1.73 m² lower and 19 ml/min per 1.73 m² higher, respectively, than median eGFR in men and women. eGFR curves by study and sex are presented in Supplementary Figure S2. Results using the RLM, CKD-EPI₂₀₀₉, and CKD-EPI₂₀₂₁ equations are presented in Supplementary Tables S4-S6 and in Supplementary Figures S3-S5. Compared with the main

Country	Region(s) or cities	Study	Ages sampled, yr ^a	Sampling frame	Sample selection	Response, %
Iceland	All	Iceland CKD study ²⁶	≥18	National health service	Inhabitants of Iceland with ≥ 1 SCr measurements available	66
Italy	Northeast	INCIPE ²⁷	≥23	General practitioner lists	Random selection of participants from 62 randomly selected practices	62
Germany	Berlin	BIS ²⁸	≥70	Individuals insured with the AOK-Nordost statutory health insurance company	Random selection of participants insured by the AOK Nordost	8.1
	Northeast	SHIP ²⁹	20–80	Population registers	Stratification based on number of residents per municipality. Age and sex stratified random sample selection per community	69
The Netherlands	Northern	Lifelines ³⁰	≥18	General practitioner lists	General practitioners invited patients to participate, and patients invited family members to participate	58–84 ^b
Norway	Central	The HUNT Study ^{31,32}	20-80	Census data	All residents in region	54
	Tromsø municipality	The Tromsø Study ³³	≥40	Population registries	Random selection of individuals from birth cohorts and population registries were invited to participate	65
Sweden	Region of Stockholm	SCREAM ³⁴	≥18	Complete health system	All Stockholm residents with a valid personal identifying number undertaking at least 1 measurement of creatinine in connection with a health care encounter	69 ^c
United Kingdom	England, Wales, and Scotland	UK Biobank ³⁵	40–70	General practitioner lists	All individuals on NHS patient registers living within a reasonable traveling distance of an assessment center	5.5

Table 1 | Description of participating studies and sampling characteristics

AOK, Allgemeine Ortskrankenkasse; BIS, The Berlin Initiative Study; CKD, chronic kidney disease; The HUNT Study, The Trøndelag Health Study; INCIPE, Initiative on Nephropathy, of relevance to public health, which is Chronic, possibly in its Initial stages, and carries a Potential risk of major clinical Endpoints; NHS, National Health Service; SCr, serum creatinine; SCREAM, Stockholm CREAtinine Measurement; SHIP, Study of Health in Pomerania. ^aStudies that collected data on all ages were asked to provide only data on those aged \geq 18 years.

^bDependent on questionnaire.

^cThis is the proportion of the sampling population that underwent creatinine testing during the study period.



Figure 1 | Flowchart illustrating the process to select healthy individuals from the total amount of individuals combined from all 9 participating studies. Red boxes indicate excluded individuals, and the green box represents the individuals used in the analysis.

results, the RLM equation showed eGFR decreasing at a similar age but provided lower eGFR values. Both the CKD- EPI_{2009} and CKD- EPI_{2021} displayed slightly higher overall eGFR values and showed a linear decrease in eGFR from the age of 20 years.

Sensitivity analyses

Results remained similar in a randomly selected subset of data reflecting a more equal contribution from each cohort (Supplementary Table S7 and Supplementary Figure S6). The median difference in the 50th percentile of eGFR values between this sensitivity analysis and the main analysis was between 2% and 3%. A larger than 5% difference was seen after the age of 87 years in men and after the age of 86 years in women. However, only 5 men and 5 women in the sensitivity analysis were aged ≥90 years. In Supplementary Table S8, we present the bias, precision, and accuracy of eGFR calculated with the EKFC equation from 1324 individuals aged 55 to 70 years who had measurements of glomerular filtration rate (GFR) available in the Renal Iohexol Clearance Survey–Follow-up, a substudy of the Tromsø Study, Norway.³⁶

DISCUSSION

Summary of findings

Our study describes sex-specific EKFC_{crea}-based eGFR reference values across the entire life span, in a large multicohort sample of healthy Europeans, and demonstrates that kidney function is not preserved with aging in healthy individuals. eGFR decreased after the age 40 years, reflecting the structure of the EKFC equation, with a slightly stronger eGFR-age association observed in women compared with men.

Measured GFR across age by sex in healthy individuals

Kidney function inevitably declines with aging, with loss of kidney function at the structural level paralleling the decline in mGFR associated with aging.^{37,38} In studies of healthy individuals or kidney donors using mGFR, kidney function decreases linearly with age,^{8,11,16,39} and in studies using nonlinear models, an initiation in decline occurs between the ages of 30 and 50 years.^{7,10,40} The Renal Iohexol Clearance Survey in Tromsø, Norway, the only longitudinal study in the general population using repeated measurements of GFR, found healthy men had a steeper GFR decline compared with women from the ages of 50 to 75 years.³⁶ Consequently, women had a lower GFR than men at the age of 50 to 65 years, whereas men had a lower GFR after. In contrast, crosssectional studies show a stronger association between age and lowering mGFR in women compared with men,^{6,9,11,41} also reported by a meta-analysis consisting of 12 cross-sectional studies in healthy potential living kidney donors.⁴² Consequently, women had lower or similar mGFR compared with men.^{8–11,42,43} Cross-sectional studies describe the distribution of GFR at a specific time point (and thus specific year of age) independent of previous kidney function, whereas longitudinal studies describe GFR over a period of time, detailing age-related kidney function trajectories.

There is some evidence that longitudinal changes in kidney function may slightly differ when using eGFR or mGFR.^{44–46} mGFR is likely best used in longitudinal studies on kidney function, and, because of its accuracy, in individuals with suspected kidney damage. Conversely, eGFR remains an important tool for describing trends at the population level because eGFR is commonly used as a first-line diagnostic tool for kidney function in primary care and clinical settings because of its accessibility and low cost.⁴⁷

Differences in eGFR values between men and women

Although cross-sectional studies on eGFR over age have similar findings to those reported from studies in mGFR,^{13,14,48} observed differences in mGFR and eGFR between men and women are likely caused by both gender (nonbiological) and sex (biological) differences. Genderrelated differences can be attributed to one's cultural and societal environment, such that men and women have differences in prevalence of risk factors or comorbidities that may impact kidney function.⁴⁹ By selecting only healthy individuals, we sought to remove the impact of disease-related gender risk factors for kidney damage, which allows for a

Table 2 | Study population characteristics for individuals who fulfill the healthy^a criteria

Variable	lceland CKD study (Iceland)	INCIPE (Italy)	BIS (Germany)	SHIP (Germany)	Lifelines (The Netherlands)	The HUNT Study (Norway)	The Tromsø Study (Norway)	SCREAM (Sweden)	UK Biobank (United Kingdom)	Total
Individuals who fulfill the healthy criteria, n (%) ^b	155,185 (71)	808 (22)	138 (7)	987 (24)	40,557 (31)	9491 (23)	4415 (23)	1,101,694 (68)	221,978 (47)	1,535,253 (60)
Healthy men, n (%) (of healthy individuals)	71,918 (46)	265 (33)	62 (45)	373 (38)	16,657 (41)	4208 (44)	2072 (47)	507,632 (46)	88,519 (40)	691,706 (45)
Age range, yr, minimum-maximum	18–105	24–93	69–97	20-80	18–90	19–96	40–91	18–107	37–70	18–107
Age, yr	41 (16)	56 (11)	78 (7)	43 (13)	39 (12)	44 (14)	52 (10)	41 (15)	54 (8)	43 (15)
eGFR by EKFC, ml/min per 1.73 m ²	92 (16)	87 (15)	66 (12)	92 (15)	95 (13)	98 (14)	89 (13)	96 (15)	89 (12)	94 (15)
eGFR by revised Lund-Malmö, ml/min per 1.73 m ²	86 (15)	82 (13)	64 (12)	86 (13)	88 (10)	92 (13)	83 (11)	89 (14)	84 (11)	88 (14)
eGFR by CKD-EPI ₂₀₀₉ , ml/min per 1.73 m ²	98 (19)	90 (15)	73 (14)	96 (17)	100 (15)	103 (16)	91 (13)	102 (18)	93 (12)	100 (18)
eGFR by CKD-EPI ₂₀₂₁ , ml/min per 1.73 m ²	101 (18)	94 (14)	78 (14)	99 (16)	103 (15)	106 (15)	95 (12)	104 (17)	96 (12)	103 (16)

BIS, The Berlin Initiative Study; CKD, chronic kidney disease; CKD-EPI, Chronic Kidney Disease Epidemiology Collaboration; eGFR, estimated glomerular filtration rate; EKFC, European Kidney Function Consortium; The HUNT Study, The Trøndelag Health Study; INCIPE, Initiative on Nephropathy, of relevance to public health, which is Chronic, possibly in its Initial stages, and carries a Potential risk of major clinical Endpoints; SCREAM, Stockholm CREAtinine Measurement; SHIP, Study of Health in Pomerania.

^aHealthy refers to individuals with (i) no history of myocardial infarction, (ii) no history of angina pectoris, (iii) no history of heart failure, (iv) no history of coronary artery disease, (v) no history of hypertension, (vi) no history of stroke, (vii) no history of cancer, (viii) no history of diabetes, (ix) no history of kidney disease, (x) body mass index \leq 30 kg/m², (xi) never smoked, (xii) no use of lipid-lowering medication or cardiac glycosides, (xiii) urine albumin-to-creatinine ratio \leq 30 mg/g, and (xiv) no use of angiotensin-converting enzyme inhibitors or angiotensin receptor blockers.

^bThis proportion of individuals who fulfill the healthy criteria is calculated from eligible individuals with no missing data.

Data are mean (SD) unless otherwise noted.



Age (yr) Figure 2 | Median estimated glomerular filtration rate (eGFR; ml/min per 1.73 m²) using the creatinine-based European Kidney Function Consortium (EKFC) equation in healthy European adult populations aged 18 to 100 years by sex. The line colors indicate sex, with blue for men and red for women. The solid line indicates the 50th percentile. The dashed lines represent the 5th (lower) and 95th

more accurate evaluation of the impact of biological sex on eGFR, although there likely still remains some residual gender bias. Sex-related hormonal differences have been hypothesized to contribute to the differences in GFR between men and women. Some have speculated that sex hormones, especially estrogen, may protect nonmenopaused women from GFR decline, potentially explaining a possible steeper decline in GFR around middle age in women compared with men.^{7,8,50} However, studies in postmenopausal women have found conflicting results, with some studies reporting hormone use to be protective against kidney function loss,⁵¹ whereas others have not found any association.⁵²⁻⁵⁴ On the other hand, studies have shown low testosterone to be associated with reduced eGFR⁵⁵ and CKD,^{56–58} but 2 studies reported low-dose testosterone supplements to delay CKD progression in hypogonadal men.^{59,60} Therefore, a lack of consensus of the effects of sex hormones on kidney function remains, and further research on the relationship between hormones and kidney function in humans is warranted.

In addition to hormonal differences, differences in eGFR between men and women may be due to non–GFR-related confounding in the calculation of creatinine estimated GFR. eGFR equations traditionally use a standard body surface area of 1.73 m^2 for normalization of GFR regardless of sex. However, body surface area averages around 2.0 m² in men and 1.8 m² in women.^{41,61} Considering this, Eriksen *et al.* found that in women, eGFR indexed to sex-specific average body surface area was 14% higher than eGFR

1082

(upper) percentiles.

indexed to 1.73 m² (99.8 vs. 86.5 ml/min).⁶¹ In men however, the difference was <1% (105.7 vs. 105.0 ml/min),⁶¹ suggesting that indexing to the same body surface area for both men and women—as is currently common practice—may induce or amplify sex differences in eGFR.^{62,63} Muscle mass also impacts creatinine production, potentially contributing to sex differences in eGFR.⁶⁴ and the trajectory of eGFR associated with aging in men and women.³⁶ To overcome this, eGFR calculated using cystatin C has been shown to estimate GFR more accurately than serum creatinine, both in individuals with low muscle mass and in the general population.^{65–67} An assessment of eGFR reference values based on cystatin C may account for muscle mass–related sex differences.

eGFR equations

There are many equations to estimate GFR, but in this study, we chose the EKFC equation for the following reasons. Although the CKD-EPI₂₀₀₉ equation has been commonly used and was recommended by the 2012 KDIGO guidelines for diagnosis of CKD,¹ its use in European populations may not be ideal, as the equation was derived from an ethnically diverse American population with a small number of elderly persons, includes a race parameter, and assumes a linear decline in kidney function with age in adulthood. This linear decline assumption explains why eGFR values in younger individuals estimated with the CKD-EPI₂₀₀₉ are higher, and more biased, than with the EKFC equation.⁶⁸ The updated CKD-EPI₂₀₂₁ equation removed the race parameter, but

Sex	Age, yr	N	2.5th	5th	eGFR, ml/min per 1.73 m ² , by percentiles							
					10th	25th	50th	75th	90th	95th	97.5th	σ^{b}
Men	20	12,806	75	78	83	90	99	107	114	119	122	12
	25	12,714	77	81	85	92	101	109	116	120	124	12
	30	14,164	77	81	85	93	101	109	116	121	125	12
	35	13,947	77	81	86	93	101	109	116	121	125	12
	40	16,284	77	80	85	92	100	108	115	119	123	12
	45	17,304	73	76	80	87	95	102	109	113	117	11
	50	15,789	69	72	76	83	90	98	104	108	112	11
	55	12,563	66	69	73	79	86	93	100	103	107	10
	60	10,647	63	66	70	76	82	89	95	99	102	10
	65	7540	60	63	66	72	79	85	91	94	97	10
	70	3324	55	58	62	68	75	81	87	91	94	10
	75	1708	50	53	57	64	71	78	84	88	91	10
	80	820	45	49	53	59	66	74	80	84	88	11
	85	436	40	44	48	55	62	70	77	81	85	11
	90	178	35	39	43	50	58	67	74	78	82	12
	95	30	30	34	38	46	55	64	71	76	80	13
	100	5	25	29	34	42	52	61	69	74	78	14
Women	20	18,069	77	81	85	93	101	109	117	121	125	12
	25	16,895	79	83	87	94	102	111	118	122	126	12
	30	17,108	79	82	87	94	102	110	118	122	126	12
	35	15,807	78	82	87	94	102	110	118	122	126	12
	40	18,059	77	81	85	92	100	108	116	120	124	12
	45	19,482	73	76	80	87	95	103	109	114	117	11
	50	18,018	69	72	76	83	90	97	104	108	111	11
	55	14,614	65	68	72	78	85	92	99	103	106	10
	60	13,205	62	65	69	75	81	88	94	98	101	10
	65	9704	58	61	65	70	77	83	89	93	96	10
	70	4297	53	56	60	66	72	79	85	89	92	10
	75	2251	48	51	55	61	68	75	81	85	88	10
	80	1241	42	46	50	56	63	71	77	81	85	11
	85	736	37	40	44	51	59	67	73	78	81	11
	90	387	31	35	39	46	55	63	70	74	78	12
	95	131	25	29	34	42	50	59	67	72	76	13
	100	28	20	25	29	38	47	56	64	69	74	14

Table 3 | eGFR percentiles by age and sex in 5-year intervals from 20 to 100 years using the EKFC_{crea} equation in healthy individuals^a and σ parameter^b at each age interval

eGFR, estimated glomerular filtration rate; EKFC_{crea}, creatinine-based European Kidney Function Consortium equation.

^aHealthy refers to individuals with (i) no history of myocardial infarction, (ii) no history of angina pectoris, (iii) no history of heart failure, (iv) no history of coronary artery disease, (v) no history of hypertension, (vi) no history of stroke, (vii) no history of cancer, (viii) no history of diabetes, (ix) no history of kidney disease, (x) body mass index \leq 30 kg/m², (xi) never smoked, (xii) no use of lipid-lowering medication or cardiac glycosides, (xiii) urine albumin-to-creatinine ratio \leq 30 mg/g, and (xiv) no use of angiotensin-converting enzyme inhibitors or angiotensin receptor blockers.

^bThis proportion of individuals who fulfill the healthy criteria is calculated from eligible individuals with no missing data. The σ parameter is a coefficient of variation representing the SD of the normal distribution.

performs worse than the CKD-EPI₂₀₀₉ in European populations.⁶⁹ More accurate estimates of GFR in European populations have been achieved with the EKFC^{68,70–73} or RLM^{74–77} equations, both derived from White European populations with a larger age range and incorporation of a nonlinear dependency on age. However, there is still no consensus on whether the EKFC equation or RLM equation has a better performance in European populations.^{78,79}Studies in predominantly European populations show bias of the EKFC equation to range between –0.9 and 3.5 ml/min per 1.73 m², ^{18,68,70} precision (the interquartile range of the absolute bias) ranging between 14.5 and 16.8 ml/min per

1.73 m²,^{18,68} and accuracy (percentage of individuals with an eGFR within 30% of the value of mGFR) ranging between 88.0% and 90.6%.^{18,70} One study in the SCREAM cohort, which constituted most of our analytical cohort, found the EKFC to have a bias of 2.7 ml/min per 1.73 m², a precision of 15.6 ml/min per 1.73 m², and an accuracy of 79.5%.⁷⁶

Application of eGFR reference values

Reference values are commonly used in medical fields to define abnormal biological values, making them important tools within clinical and public health settings.⁸⁰ eGFR reference values play an important role in the field of

nephrology and throughout the life course of a kidney. A study assessing potential kidney donors by age and eGFR suggested that age-calibrated eGFR assessments could improve the efficiency of living kidney donor selection.^{81,82} Others have suggested that reference values could improve the detection of abnormal kidney function, and be especially relevant for screening and management of risk factors, and referral to specialist care. This is especially important in younger adults, as they are less likely to be referred when their eGFR is below the expected normal value compared with older age groups.⁸³ For instance, as the lower 5th percentile of eGFR in healthy people aged <50 years lies between 75 and 85 ml/min per 1.73 m², clinicians may opt to test for other signs of kidney damage, such as albuminuria, before eGFR reaches the KDIGO threshold of 60 ml/min per 1.73 m². This situation was explored in a randomized vignette study, which found that providing general practitioners (GPs) with age-specific eGFR reference values significantly increased the proportion of GPs identifying a clinical problem in younger adults with reduced kidney function from 47% to 84% of GPs, even when eGFR was >60 ml/min per 1.73 m^{2.84} A follow-up study found that 89% of GPs had positive views of the incorporation of eGFR reference values in clinical settings (the remaining GPs had neutral views) and described it as "easy to use" and a "valuable tool" to assess kidney function by age.^{85(p.3)}

The observation of declining eGFR with age has also prompted some people to suggest the use of an age-adapted CKD definition to account for "kidney aging."^{2,86-88} Using the reference values established in this study, it is possible to develop a continuous age-adapted CKD definition using the age- and sex-specific lower 5th percentile of the eGFR reference value distribution.² Furthermore, there is a large proportion of elderly individuals needing kidney replacement therapy who do not have an identified cause of CKD. An insufficient diagnostic workup may contribute to this observation, or there may be no specific cause and these individuals represent one extreme of kidney aging that may be labeled "accelerated kidney aging" because of the interaction of genetic predisposition,⁸⁹ nephron number at birth,^{90–92} and environment.⁹³ The 5th percentile of the eGFR reference value distribution in otherwise healthy individuals may help to develop the conceptual framework to define accelerated kidney aging.

Strengths and limitations

Although eGFR may provide a biased estimate of GFR at the individual level compared with mGFR, the bias between eGFR and mGFR at a population level within a healthy, homogeneous adult population should be minimal.⁹⁴ Our cross-sectional study design allows us to describe normal and abnormal eGFR distribution across an age spectrum, but kidney function trajectories cannot be extrapolated from our results. We sampled from general population studies across multiple European countries, but the generalizability of these results in Europe should be considered with caution, especially as no cohorts from Eastern Europe were able to participate. Also, the contribution of each participating study (sample size) was not

equal, but our sensitivity analysis showed cohort sample sizes had little impact on the results. Our healthy definition, which included 14 criteria, should identify individuals without comorbid conditions or certain CKD risk factors. Although these variables had varying collection methods among the cohorts, with some missing variables at the cohort level, we believe that most individuals who would not fulfill the healthy criteria were captured by at least 1 of the other exclusion criteria. For instance, individuals with elevated albumin-to-creatinine ratio, a variable missing for most of the cohort, may be captured by the variable kidney disease, available in all cohorts. Similarly, missing data on angiotensin-converting enzyme inhibitor/ angiotensin receptor blocker use may be captured by variables on cardiovascular disease or kidney disease. Nonetheless, the consequence of missing these critical variables to assess kidney health is that individuals with subclinical or unrecognized kidney disease may still be present in our healthy population. Residual bias may also remain as other relevant comorbidities and lifestyle factors impacting kidney health, such as diet and genetic factors, were not included in our healthy criteria. Results from our study may be impacted by survivor bias, as only the healthiest individuals are likely to survive to old age, so our estimates of GFR in especially older healthy individuals may be slightly inflated. Additionally, by defining normal and abnormal values from a healthy population based purely on statistical distributions, we assume that 5% of otherwise healthy individuals had an abnormal eGFR.

Conclusion

Our results from >1.5 million healthy individuals from a multinational, contemporary, adult European general population suggest that kidney function is not preserved with aging in healthy individuals. Creatinine-based eGFR is lower at higher ages with a slightly steeper eGFR-age decrease in women than men, leading to a lower eGFR in women compared with men after the age of ~60 years. The eGFR reference values described here provide important information relevant to understanding kidney health in relation to aging by sex and can contribute to the debate on how CKD should be defined and handled throughout the life course.

DISCLOSURE

AO has received grants from Sanofi; has consulted or received speaker fees or travel support from Adviccene, Alexion, Astellas, AstraZeneca, Amicus, Amgen, Boehringer Ingelheim, Fresenius Medical Care, GSK, Bayer, Sanofi-Genzyme, Sobi, Menarini, Mundipharma, Kyowa Kirin, Lilly, Freeline, Idorsia, Chiesi, Otsuka, Novo-Nordisk, Sysmex, Vifor Fresenius Medical Care Renal Pharma, and Spafarma; and is director of the Catedra UAM-AstraZeneca of chronic kidney disease and electrolytes. He has stock in Telara Farma. ES receives honoraria from the National Kidney Diseases and from AstraZeneca for consulting. NE receives honoraria from Bayer AG. All the other authors declared no competing interests.

DATA STATEMENT

The data used in this study are not available because of data protection laws and regulations.

ACKNOWLEDGMENTS

This research has been conducted using the UK Biobank Resource under application number 109166.

FUNDING STATEMENT

The European Renal Association (ERA) Registry is funded by the ERA. This article was written by Astley et al. on behalf of the ERA Registry, which is an official body of ERA. The Berlin Initiative Study (BIS) is funded by the KfH Foundation for Preventive Medicine and the DDnÄ Institut für Disease Management. Stockholm CREAtinine Measurement (SCREAM) is supported by the Swedish Research Council, the Swedish Heart and Lung Foundation, and Region Stockholm (ALF Medicin). AO's research is funded by Instituto de Salud Carlos III FIS/Fondos FEDER RICORS program to RICORS2040 (RD21/0005/0001) funded by European Union-NextGenerationEU, Mecanismo para la Recuperación y la Resiliencia, and SPACKDc PMP21/00109, COST Action PERMEDIK CA21165, supported by COST (European Cooperation in Science and Technology). This work was supported by the PREVENTCKD Consortium (Project identifier: 101101220, Programme: EU4H, DG/Agency: HADEA) and by the KitNewCare Project (Project identifier: 101137054, Call: HORIZON-HLTH-2023-CARE-04, Programme: HORIZON, DG/Agency: HADEA).

AUTHOR CONTRIBUTIONS

KJJ, VSS, AO, SH, GG, NCC, and MEA conceived and designed the analysis; SH, GG, J-JC, NE, BOE, A-LF, PMF, OSI, TI, AJJ, KARL, TM, ES, and SS collected or contributed data; MEA and NCC performed the analysis; and MEA, KJJ, VSS, AO, SH, GG, NCC, J-JC, NE, BOE, A-LF, PMF, OSI, TI, AJJ, KARL, TM, ES, and SS performed writing and reviewing.

Supplementary material is available online at www.kidneyinternational.org.

REFERENCES

- 1. Stevens PE, Levin A. Evaluation and management of chronic kidney disease: synopsis of the Kidney Disease: Improving Global Outcomes 2012 clinical practice guideline. *Ann Intern Med.* 2013;158:825–830.
- Delanaye P, Jager KJ, Bökenkamp A, et al. CKD: a call for an age-adapted definition. J Am Soc Nephrol. 2019;30:1785–1805.
- Delanaye P, Schaeffner E, Ebert N, et al. Normal reference values for glomerular filtration rate: what do we really know? *Nephrol Dial Transplant*. 2012;27:2664–2672.
- de Onis M, Onyango AW, Borghi E, et al. Development of a WHO growth reference for school-aged children and adolescents. *Bull World Health Organ.* 2007;85:660–667.
- Mangseth H, Sikkeland LIB, Durheim MT, et al. Comparison of different reference values for lung function: implications of inconsistent use among centers. *BMC Pulmonary Med.* 2023;23:137.
- Pottel H, Delanaye P, Weekers L, et al. Age-dependent reference intervals for estimated and measured glomerular filtration rate. *Clin Kidney J*. 2017;10:545–551.
- Poggio ED, Rule AD, Tanchanco R, et al. Demographic and clinical characteristics associated with glomerular filtration rates in living kidney donors. *Kidney Int.* 2009;75:1079–1087.
- 8. Berg UB. Differences in decline in GFR with age between males and females: reference data on clearances of inulin and PAH in potential kidney donors. *Nephrol Dial Transplant*. 2006;21:2577–2582.
- 9. Rule AD, Gussak HM, Pond GR, et al. Measured and estimated GFR in healthy potential kidney donors. *Am J Kidney Dis.* 2004;43:112–119.
- Grewal GS, Blake GM. Reference data for 51Cr-EDTA measurements of the glomerular filtration rate derived from live kidney donors. *Nucl Med Commun.* 2005;26:61–65.
- 11. Fenton A, Montgomery E, Nightingale P, et al. Glomerular filtration rate: new age- and gender- specific reference ranges and thresholds for living kidney donation. *BMC Nephrol.* 2018;19:336.

- Schaeffner ES, Ebert N, Kuhlmann MK, et al. Age and the course of GFR in persons aged 70 and above. *Clin J Am Soc Nephrol.* 2022;17:1119–1128.
- **13.** Herold JM, Wiegrebe S, Nano J, et al. Population-based reference values for kidney function and kidney function decline in 25- to 95-year-old Germans without and with diabetes. *Kidney Int.* 2024;106:699–711.
- Wetzels JF, Kiemeney LA, Swinkels DW, et al. Age- and gender-specific reference values of estimated GFR in Caucasians: the Nijmegen Biomedical Study. *Kidney Int*. 2007;72:632–637.
- **15.** Hannemann A, Friedrich N, Dittmann K, et al. Age- and sex-specific reference limits for creatinine, cystatin C and the estimated glomerular filtration rate. *Clin Chem Lab Med.* 2011;50:919–926.
- Eriksen BO, Palsson R, Ebert N, et al. GFR in healthy aging: an individual participant data meta-analysis of iohexol clearance in European population-based cohorts. J Am Soc Nephrol. 2020;31:1602–1615.
- Brück K, Stel VS, Gambaro G, et al. CKD prevalence varies across the European general population. J Am Soc Nephrol. 2016;27:2135–2147.
- Pottel H, Björk J, Courbebaisse M, et al. Development and validation of a modified full age spectrum creatinine-based equation to estimate glomerular filtration rate. Ann Intern Med. 2021;174:183–191.
- Kidney Disease: Improving Global Outcomes (KDIGO) CKD Work Group. KDIGO 2024 Clinical Practice Guideline for the Evaluation and Management of Chronic Kidney Disease. *Kidney Int*. 2024;105(45):S117–S314.
- 20. Björk J, Grubb A, Sterner G, Nyman U. Revised equations for estimating glomerular filtration rate based on the Lund-Malmö Study cohort. *Scand J Clin Lab Invest*. 2011;71:232–239.
- 21. Levey AS, Stevens LA, Schmid CH, et al. A new equation to estimate glomerular filtration rate. *Ann Intern Med.* 2009;150:604–612.
- Inker LA, Eneanya ND, Coresh J, et al. New creatinine- and cystatin Cbased equations to estimate GFR without race. N Engl J Med. 2021;385: 1737–1749.
- 23. Stasinopoulos DM, Rigby RA. Generalized additive models for location scale and shape (GAMLSS) in R. J Stat Software. 2007;23:1–46.
- 24. Rigby RA, Stasinopoulos DM. Generalized additive models for location, scale and shape, (with discussion). *Appl Stat.* 2005;54:507–554.
- 25. R: A Language and Environment for Statistical Computing. 2023. Accessed December 20, 2024. https://www.r-project.org/
- Jonsson AJ, Lund SH, Eriksen BO, et al. The prevalence of chronic kidney disease in Iceland according to KDIGO criteria and age-adapted estimated glomerular filtration rate thresholds. *Kidney Int*. 2020;98:1286– 1295.
- 27. Gambaro G, Yabarek T, Graziani MS, et al. Prevalence of CKD in northeastern Italy: results of the INCIPE study and comparison with NHANES. *Clin J Am Soc Nephrol.* 2010;5:1946–1953.
- Ebert N, Jakob O, Gaedeke J, et al. Prevalence of reduced kidney function and albuminuria in older adults: the Berlin Initiative Study. *Nephrol Dial Transplant*. 2017;32:997–1005.
- **29.** Völzke H. Study of Health in Pomerania (SHIP). *Bundesgesundheitsblatt Gesundheitsforschung Gesundheitsschutz*. 2012;55:790–794.
- Scholtens S, Smidt N, Swertz MA, et al. Cohort profile: LifeLines, a threegeneration cohort study and biobank. *Int J Epidemiol*. 2015;44:1172–1180.
- **31.** Åsvold BO, Langhammer A, Rehn TA, et al. Cohort profile update: the HUNT study, Norway. *Int J Epidemiol.* 2023;52:e80–e91.
- Gustad LT, Myklebust TÅ, Bjerkeset O, et al. Anxiety and depression symptoms, albuminuria and risk of acute myocardial infarction in the Norwegian HUNT cohort study. *BMC Cardiovasc Disord*. 2022;22:472.
- **33.** Hopstock LA, Grimsgaard S, Johansen H, et al. The seventh survey of the Tromsø Study (Tromsø7) 2015-2016: study design, data collection, attendance, and prevalence of risk factors and disease in a multipurpose population-based health survey. *Scand J Public Health*. 2022;50:919–929.
- **34.** Carrero JJ, Elinder CG. The Stockholm CREAtinine Measurements (SCREAM) project: fostering improvements in chronic kidney disease care. *J Intern Med.* 2022;291:254–268.
- 35. Allen N, Sudlow C, Downey P, et al. UK Biobank: current status and what it means for epidemiology. *Health Policy Technol.* 2012;1:123–126.
- **36.** Melsom T, Norvik JV, Enoksen IT, et al. Sex differences in age-related loss of kidney function. J Am Soc Nephrol. 2022;33:1891–1902.
- Denic A, Lieske JC, Chakkera HA, et al. The substantial loss of nephrons in healthy human kidneys with aging. J Am Soc Nephrol. 2017;28:313–320.
- Denic A, Rule AD, Glassock RJ. Healthy and unhealthy aging on kidney structure and function: human studies. *Curr Opin Nephrol Hypertens*. 2022;31:228–234.
- **39.** Rule AD, Amer H, Cornell LD, et al. The association between age and nephrosclerosis on renal biopsy among healthy adults. *Ann Intern Med.* 2010;152:561–567.

- 40. Delanaye P, Gaillard F, van der Weijden J, et al. Age-adapted percentiles of measured glomerular filtration in healthy individuals: extrapolation to living kidney donors over 65 years. *Clin Chem Lab Med*. 2022;60:401–407.
- **41.** Peters AM, Perry L, Hooker CA, et al. Extracellular fluid volume and glomerular filtration rate in 1878 healthy potential renal transplant donors: effects of age, gender, obesity and scaling. *Nephrol Dial Transplant*. 2012;27:1429–1437.
- **42.** Pottel H, Hoste L, Yayo E, Delanaye P. Glomerular filtration rate in healthy living potential kidney donors: a meta-analysis supporting the construction of the full age spectrum equation. *Nephron.* 2017;135:105–119.
- 43. Blake GM, Sibley-Allen C, Hilton R, et al. Glomerular filtration rate in prospective living kidney donors. *Int Urol Nephrol.* 2013;45:1445–1452.
- de Boer IH, Sun W, Cleary PA, et al. Longitudinal changes in estimated and measured GFR in type 1 diabetes. J Am Soc Nephrol. 2014;25:810– 818.
- **45.** Pottel H, Delay A, Maillard N, et al. 20-Year longitudinal follow-up of measured and estimated glomerular filtration rate in kidney transplant patients. *Clin Kidney J.* 2021;14:909–916.
- **46.** Ku E, Xie D, Shlipak M, et al. Change in measured GFR versus eGFR and CKD outcomes. *J Am Soc Nephrol*. 2016;27:2196–2204.
- 47. Inker LA, Titan S. Measurement and estimation of GFR for use in clinical practice: core curriculum 2021. *Am J Kidney Dis.* 2021;78:736–749.
- Hallan S, Astor B, Lydersen S. Estimating glomerular filtration rate in the general population: the second Health Survey of Nord-Trondelag (HUNT II). Nephrol Dial Transplant. 2006;21:1525–1533.
- 49. Carrero JJ. Gender differences in chronic kidney disease: underpinnings and therapeutic implications. *Kidney Blood Press Res.* 2010;33:383–392.
- Farahmand M, Ramezani Tehrani F, et al. Endogenous estrogen exposure and chronic kidney disease; a 15-year prospective cohort study. BMC Endocr Disord. 2021;21:155.
- 51. Schopick EL, Fisher ND, Lin J, et al. Post-menopausal hormone use and albuminuria. *Nephrol Dial Transplant*. 2009;24:3739–3744.
- 52. Monster TBM, Janssen WMT, de Jong PE, et al. Oral contraceptive use and hormone replacement therapy are associated with microalbuminuria. *Arch Intern Med.* 2001;161:2000–2005.
- **53.** Ahmed SB, Culleton BF, Tonelli M, et al. Oral estrogen therapy in postmenopausal women is associated with loss of kidney function. *Kidney Int.* 2008;74:370–376.
- Kim C, Saran R, Hood M, et al. Changes in kidney function during the menopausal transition: the Study of Women's Health Across the Nation (SWAN) - Michigan site. *Menopause*. 2020;27:1066–1069.
- 55. Kurita N, Horie S, Yamazaki S, et al. Low testosterone levels and reduced kidney function in Japanese adult men: the Locomotive Syndrome and Health Outcome in Aizu Cohort Study. J Am Med Directors Assoc. 2016;17: 371.e1–371.e6.
- Amiri M, Ramezani Tehrani F, Rahmati M, et al. Low serum testosterone levels and the incidence of chronic kidney disease among male adults: a prospective population-based study. *Andrology*. 2020;8:575–582.
- Zhao JV, Schooling CM. Sex-specific associations of sex hormone binding globulin with CKD and kidney function: a univariable and multivariable Mendelian randomization study in the UK Biobank. J Am Soc Nephrol. 2021;32:686–694.
- Zhao JV, Schooling CM. The role of testosterone in chronic kidney disease and kidney function in men and women: a bi-directional Mendelian randomization study in the UK Biobank. *BMC Med.* 2020;18:122.
- Sharma R, Oni O, Wiegmann P, et al. Testosterone replacement therapy (TRT) is associated with delayed progression of chronic kidney disease: a retrospective analysis of testosterone normalization in US veterans. *Ann Nephrol.* 2020;5:51–59.
- **60.** Alwani M, Al-Zoubi RM, Al-Qudimat A, et al. The impact of long-term testosterone therapy (TTh) in renal function (RF) among hypogonadal men: an observational cohort study. *Ann Med Surg.* 2021;69:102748.
- 61. Eriksen BO, Melsom T, Mathisen UD, et al. GFR normalized to total body water allows comparisons across genders and body sizes. *J Am Soc Nephrol*. 2011;22:1517–1525.
- **62.** Korhonen PE, Ekblad MO, Kautiainen H, Mäkelä S. Renal hyperfiltration revisited—role of the individual body surface area on mortality. *Eur J Intern Med.* 2023;114:101–107.
- **63.** Redal-Baigorri B, Rasmussen K, Heaf JG. Indexing glomerular filtration rate to body surface area: clinical consequences. *J Clin Lab Anal*. 2014;28: 83–90.
- 64. Nankivell BJ, Nankivell LFJ, Elder GJ, Gruenewald SM. How unmeasured muscle mass affects estimated GFR and diagnostic inaccuracy. *eClinicalMedicine*. 2020;29-30:100662.

- **65.** Baxmann AC, Ahmed MS, Marques NC, et al. Influence of muscle mass and physical activity on serum and urinary creatinine and serum cystatin C. *Clin J Am Soc Nephrol.* 2008;3:348–354.
- **66.** Pottel H, Björk J, Rule AD, et al. Cystatin C–based equation to estimate GFR without the inclusion of race and sex. *N Engl J Med.* 2023;388:333–343.
- 67. Groothof D, Post A, Polinder-Bos HA, et al. Muscle mass and estimates of renal function: a longitudinal cohort study. *J Cachexia Sarcopenia Muscle*. 2022;13:2031–2043.
- **68.** Delanaye P, Vidal-Petiot E, Björk J, et al. Performance of creatinine-based equations to estimate glomerular filtration rate in White and Black populations in Europe, Brazil and Africa. *Nephrol Dial Transplant*. 2023;38: 106–118.
- **69.** Gansevoort RT, Anders HJ, Cozzolino M, et al. What should European nephrology do with the new CKD-EPI equation? *Nephrol Dial Transplant*. 2023;38:1–6.
- Inker LA, Tighiouart H, Adingwupu OM, et al. CKD-EPI and EKFC GFR estimating equations: performance and other considerations for selecting equations for implementation in adults. *J Am Soc Nephrol.* 2023;34:1953–1964.
- Munch PV, Heide-Jørgensen U, Jensen SK, et al. Performance of the racefree CKD-EPI creatinine-based eGFR equation in a Danish cohort with measured GFR. *Clin Kidney J.* 2023;16:2728–2737.
- 72. Pottel H, Cavalier E, Björk J, et al. Standardization of serum creatinine is essential for accurate use of unbiased estimated GFR equations: evidence from three cohorts matched on renal function. *Clin Kidney J*. 2022;15:2258–2265.
- **73.** Delanaye P, Björk J, Courbebaisse M, et al. Performance of creatininebased equations to estimate glomerular filtration rate with a methodology adapted to the context of drug dosage adjustment. *Br J Clin Pharmacol.* 2022;88:2118–2127.
- 74. Nyman U, Grubb A, Larsson A, et al. The revised Lund-Malmö GFR estimating equation outperforms MDRD and CKD-EPI across GFR, age and BMI intervals in a large Swedish population. *Clin Chem Lab Med*. 2014;52:815–824.
- **75.** Björk J, Jones I, Nyman U, Sjöström P. Validation of the Lund-Malmö, Chronic Kidney Disease Epidemiology (CKD-EPI) and Modification of Diet in Renal Disease (MDRD) equations to estimate glomerular filtration rate in a large Swedish clinical population. *Scand J Urol Nephrol.* 2012;46: 212–222.
- **76.** Fu EL, Levey AS, Coresh J, et al. Accuracy of GFR-estimating equations based on creatinine, cystatin C or both in routine care. *Nephrol Dial Transplant*. 2024;39:694–706.
- 77. Björk J, Grubb A, Sterner G, et al. Performance of GFR estimating equations stratified by measured or estimated GFR: implications for interpretation. *Am J Kidney Dis.* 2015;66:1107–1108.
- 78. Nyman U, Björk J, Berg U, et al. The modified CKiD study estimated GFR equations for children and young adults under 25 years of age: performance in a European multicenter cohort. *Am J Kidney Dis.* 2022;80: 807–810.
- 79. Beridze G, Vetrano DL, Marengoni A, et al. Concordance and discrepancies among 5 creatinine-based equations for assessing estimated glomerular filtration rate in older adults. *JAMA Network Open*. 2023;6:e234211–e.
- Boyd JC. Defining laboratory reference values and decision limits: populations, intervals, and interpretations. *Asian J Androl.* 2010;12:83–90.
- Gaillard F, Courbebaisse M, Kamar N, et al. The age-calibrated measured glomerular filtration rate improves living kidney donation selection process. *Kidney Int.* 2018;94:616–624.
- Andrews PA, Burnapp L. British Transplantation Society/Renal Association UK guidelines for living donor kidney transplantation 2018: summary of updated guidance. *Transplantation*. 2018;102:e307.
- Junayd H, Nicholas G, Ayub A, et al. Associations between modest reductions in kidney function and adverse outcomes in young adults: retrospective, population based cohort study. *BMJ*. 2023;381: e075062.
- **84.** Guppy M, Glasziou P, Beller E, et al. Kidney trajectory charts to assist general practitioners in the assessment of patients with reduced kidney function: a randomised vignette study. *BMJ Evid Based Med.* 2022;27:288–295.
- **85.** Guppy M, Bowles EJ, Glasziou P, Doust J. Use of kidney trajectory charts as an adjunct to chronic kidney disease guidelines- a qualitative study of general practitioners. *PLoS One*. 2024;19:e0305605.
- Glassock RJ, Delanaye P, Rule AD. Should the definition of CKD be changed to include age-adapted GFR criteria? YES. *Kidney Int.* 2020;97: 34–37.

- Benghanem Gharbi M, Elseviers M, Zamd M, et al. Chronic kidney disease, hypertension, diabetes, and obesity in the adult population of Morocco: how to avoid "over"- and "under"-diagnosis of CKD. *Kidney Int.* 2016;89:1363–1371.
- 88. Rovin BH. Do kidneys grow old gracefully? Kidney Int. 2020;97:40-41.
- **89.** Gorski M, Tin A, Garnaas M, et al. Genome-wide association study of kidney function decline in individuals of European descent. *Kidney Int.* 2015;87:1017–1029.
- **90.** Hughson M, Farris AB, Douglas-Denton R, et al. Glomerular number and size in autopsy kidneys: the relationship to birth weight. *Kidney Int*. 2003;63:2113–2122.
- 91. Vikse BE, Irgens LM, Leivestad T, et al. Low birth weight increases risk for end-stage renal disease. J Am Soc Nephrol. 2008;19:151–157.
- **92.** Sanderson KR, Wekon-Kemeni C, Charlton JR. From premature birth to premature kidney disease: does accelerated aging play a role? *Pediatr Nephrol.* 2024;39:2001–2013.
- **93.** Yu C-C, Lin J-L, Lin-Tan D-T. Environmental exposure to lead and progression of chronic renal diseases: a four-year prospective longitudinal study. *J Am Soc Nephrol.* 2004;15:1016–1022.
- Shafi T, Zhu X, Lirette ST, et al. Quantifying individual-level inaccuracy in glomerular filtration rate estimation. *Ann Intern Med.* 2022;175:1073– 1082.